Reviewer #1:

Q.1 “the specific mechanism of beneficial gut microbiota in improving metabolic syndrome has not been explained in detail. It would be better if we could elaborate on the mechanism in detail.”

Author Response: Thank you for this recommendation. Beneficial gut microbiota produce short-chain fatty acids (SCFAs), which have an effect through tissue-specific mechanisms that a protective against metabolic syndrome by improving insulin sensitivity, such as the SCFA propionate promoting the release of glucagon like peptide-1. Additionally, metabolic syndrome is associated with inflammation, and maintenance of the gut barrier is important for preventing pro-inflammatory bacterial products from crossing into the systemic circulation. Beneficial gut microbiota promote the gut barrier through SCFA production.

Reviewer #2:

Q.1. “reduction in the Firmicutes to Bacteroidetes ratio, what would be the effect”?

Author Response: Firmicutes to Bacteroidetes ratio has been considered the hallmark of obesity and the reduction in the ratio is associated with improvement in obesity. We have added references to the article.

References:

Q.2. “decrease in Enterococci and Enterobacteriaceae”?

Author Response: A decrease in Enterococci and Enterobacteriaceae has been shown to decrease the risk for inflammatory bowel disease. We have added the reference to the article.

Reference:

“Increase in Faecalibacterium prausnitzii”?

Author Response: Increase in Faecalibacterium prausnitzii has anti-inflammatory properties. We have added the reference to the article.

Reference:
“Increase in Bifidobacteria”?

Author Response: Increase in *Bifidobacteria*, which is one of the most commonly used probiotic promotes gut health, decrease expression of inflammatory cytokines and improve insulin sensitivity. We have added the reference to the article.

Reference:


“with a decrease in Bacteroides”?

Author Response: Decrease in Bacteroides is beneficial to humans since it is a pathogen commonly found anerobic infection and causes significant antibiotic resistance. We have added the reference to the article.

Reference:


Q.3. “This statement needs reference: It is known that obesity, T2DM and CVD are caused or worsened by multiple factors like genetic predisposition, environmental factors, unhealthy high calorie diet, and sedentary lifestyle”

Author Response: Thank you for this recommendation. We have added the reference to the article.

References:


Q.4. “An increased production of SCFA has been found in obesity and decreased production of butyrate and propionate is seen in T2DM. In mice studies, butyrate was shown to be associated with increased production of Lachnospiraceae and Proteobacteria and decreased production of Clostridiaceae. SCFA can reduce obesity, so why is it contrary to the results of the latter experiment.”?

Author Response: Thank you for pointing out the discrepancy. There is typo in the first sentence, and we have rectified it to say, “Increased production of *acetate* has been found in obesity.......”
“And butyrate and propionate are produced by Postbiotics”?

Author response: The SCFA’s, butyrate and propionate are among the metabolites that are produced from the inactive microbe aka postbiotic.

Q.5. This statement needs reference: Various studies in animals have shown their benefits in improving gut microbiome composition.

Author Response: Thank you for this recommendation. The below references have been included.

Ref:

Q.7. “Another study with 40 participants with insulin resistance were placed in a double-blind trial and given either Akkermansia muciniphila or a placebo, and the study showed reduction in inflammatory markers and improved insulin sensitivity in the Akkermansia muciniphila group. What is this sentence trying to say at the end? I think it is not a summary of this article and should be placed at the front”.

Author Response: Thank you for pointing this out. We have made the necessary change by moving the sentence to the appropriate place in the article.

Q.8. “The table can be further refined, for example, what are the effects and what can be done through what mechanism”.

Author Response: Thank you for this recommendation. We have edited Table 2 to include the above recommendations.

Q.9. “Figure 2, 3, 4 in The beneficial effects of prebiotics, postbiotics, probiotics on the gut microbiome consistent, what are the different differences in their effects, compared to the common use or which one is the best, in order to be more intuitive, and then have a I think it is necessary to have a longitudinal comparison of the figure”.

Author Response: Thank you for this recommendation. We have included the differences in their effects and have included a longitudinal comparison of the figure.

EDITORIAL OFFICE’S COMMENTS

“The author(s) must include the keyword “Diabetes” in the keywords list”

Author Response: Thank you for this recommendation. We have included the keyword “Diabetes” in the keyword list.

“
“Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor”.

Author Response: Yes, we have followed the above suggestions.

“Please check and confirm whether the figures are original”

Author Response: We have added the required information on the bottom right-hand side of the picture in PowerPoint.